## A randomised, double-blind, placebocontrolled trial of repeated nebulisation of non-viral cystic fibrosis transmembrane conductance regulator (*CFTR*) gene therapy in patients with cystic fibrosis

Eric WFW Alton,<sup>1\*†</sup> David K Armstrong,<sup>2</sup> Deborah Ashby,<sup>1</sup> Katie J Bayfield,<sup>1</sup> Diana Bilton,<sup>1,3</sup> Emily V Bloomfield,<sup>1</sup> A Christopher Boyd,<sup>2†</sup> June Brand,<sup>2</sup> Ruaridh Buchan,<sup>4</sup> Roberto Calcedo,<sup>5</sup> Paula Carvelli,<sup>1</sup> Mario Chan,<sup>1</sup> Seng H Cheng,<sup>6</sup> David S Collie,<sup>7</sup> Steve Cunningham,<sup>8†</sup> Heather E Davidson,<sup>2</sup> Gwyneth Davies,<sup>1</sup> Jane C Davies, <sup>1,3†</sup> Lee A Davies, <sup>9</sup> Maria H Dewar, <sup>4</sup> Ann Doherty,<sup>2</sup> Jackie Donovan,<sup>1,3</sup> Natalie S Dwyer,<sup>1</sup> Hala I Elgmati,<sup>2</sup> Rosanna F Featherstone,<sup>1</sup> Jemyr Gavino,<sup>1</sup> Sabrina Gea-Sorli,<sup>1</sup> Duncan M Geddes,<sup>1,3</sup> James SR Gibson,<sup>2</sup> Deborah R Gill,<sup>9†</sup> Andrew P Greening,<sup>4</sup> Uta Griesenbach,<sup>1†</sup> David M Hansell,<sup>1,3</sup> Katharine Harman,<sup>1</sup> Tracy E Higgins,<sup>1†</sup> Samantha L Hodges,<sup>1</sup> Stephen C Hyde,<sup>9†</sup> Laura Hyndman,<sup>2</sup> J Alastair Innes,<sup>4†</sup> Joseph Jacob,<sup>1,3</sup> Nancy Jones,<sup>1,3</sup> Brian F Keogh,<sup>1,3</sup> Maria P Limberis,<sup>5</sup> Paul Lloyd-Evans,<sup>10</sup> Alan W Maclean,<sup>2</sup> Michelle C Manvell,<sup>1</sup> Dominique McCormick,<sup>9</sup> Michael McGovern,<sup>2</sup> Gerry McLachlan,<sup>7</sup> Cuixiang Meng,<sup>1</sup> M Angeles Montero,<sup>1,3</sup> Hazel Milligan,<sup>4</sup> Laura J Moyce,<sup>9</sup> Gordon D Murray,<sup>11†</sup> Andrew G Nicholson,<sup>1,3</sup> Tina Osadolor,<sup>1,3</sup> Javier Parra-Leiton,<sup>2</sup> David J Porteous,<sup>2†</sup>

Ian A Pringle,<sup>9</sup> Emma K Punch,<sup>1</sup> Kamila M Pytel,<sup>1</sup> Alexandra L Quittner,<sup>12</sup> Gina Rivellini,<sup>1</sup> Clare J Saunders,<sup>1</sup> Ronald K Scheule,<sup>6</sup> Sarah Sheard,<sup>1,3</sup> Nicholas J Simmonds,<sup>1,3</sup> Keith Smith,<sup>10</sup> Stephen N Smith,<sup>1</sup> Najwa Soussi,<sup>1</sup> Samia Soussi,<sup>1</sup> Emma J Spearing,<sup>1</sup> Barbara J Stevenson,<sup>2</sup> Stephanie G Sumner-Jones,<sup>9</sup> Minna Turkkila,<sup>1</sup> Rosa P Ureta,<sup>1</sup> Michael D Waller,<sup>1</sup> Marguerite Y Wasowicz,<sup>1</sup> James M Wilson<sup>5</sup> and Paul Wolstenholme-Hogg<sup>6</sup> on behalf of the UK Cystic Fibrosis Gene Therapy Consortium

<sup>1</sup>Department of Gene Therapy, Imperial College London, London, UK <sup>2</sup>Medical Genetics, Centre for Genomic and Experimental Medicine, University of Edinburgh, Edinburgh, UK

- <sup>3</sup>Royal Brompton and Harefield NHS Foundation Trust, London, UK <sup>4</sup>Western General Hospital, Edinburgh, UK
- <sup>5</sup>Gene Therapy Program, Department of Pathology and Laboratory Medicine, University of Pennsylvania, Philadelphia, PA, USA
- <sup>6</sup>Genzyme, a Sanofi Company, Framingham, MA, USA
- <sup>7</sup>The Roslin Institute and R(D)SVS, University of Edinburgh, Edinburgh, UK <sup>8</sup>Royal Hospital for Sick Children, Edinburgh, UK
- <sup>9</sup>Gene Medicine Research Group, Nuffield Division Of Clinical Laboratory Sciences, University of Oxford, Oxford, UK
- <sup>10</sup>NHS Blood and Transplant, Bristol, UK

<sup>11</sup>Centre for Population Health Sciences, University of Edinburgh, Edinburgh, UK <sup>12</sup>Department of Psychology, University of Miami, Miami, FL, USA

\*Corresponding author †Senior authors Declared competing interests of authors: All authors report grants from the National Institute for Health Research, grants from Cystic Fibrosis Trust, grants from Just Gene Therapy, grants from Medicor Foundation and report that Genzyme, a Sanofi company, manufactured and provided Lipid 67. Ian A Pringle also has a patent WO200711062 issued. Ronald K Scheule also reports other funding from Genzyme, a Sanofi company, outside the submitted work and has a patent US5783565 issued, a patent US5840710 issued, and a patent US5935936 issued. Michelle C Manvell also reports grants from Royal Brompton Biomedical Research Unit (National institute for Health Research funds), outside the submitted work. Steve Cunningham also reports personal fees from Gilead, outside the submitted work and is a principle investigator for Vertex Pharmaceuticals studies in children with cystic fibrosis. Nicholas J Simmonds also reports personal fees from cystic fibrosis Adherence Steering Committee, personal fees from Vertex advisory board, personal fees from Pharmaxis advisory board and lecture fee, personal fees from Gilead advisory boards and lecture fee, personal fees from Eumedica lecture fee, personal fees from Forest Laboratories honorarium, outside the submitted work. Eric WFW Alton also has a patent WO2013061091 pending. Christopher Boyd also has a patent WO2013061091 pending. Alexandra L Quittner is a member of the advisory board for Genentech Inc., reports consulting for AbbVie Pharmaceuticals, grants from Cystic Fibrosis Foundation, grants from Vertex Pharmaceuticals and grants from Novartis Pharmaceuticals, outside the submitted work. Uta Griesenbach also has a patent WO2013061091 pending, and a patent European Patent Application Number 12784648.3 pending. Seng H Cheng is an employee of Genzyme, a Sanofi Company and also has a patent US5650096 issued, a patent US5747471 issued, and a patent US5840710 issued. David J Porteous also has a patent WO2013061091 pending. J Alastair Innes also has a patent WO2013061091 pending. Jane C Davies also reports fees paid to employing institution from Vertex for her role as clinical trials lead, educational meetings and advisory board participation, fees paid to employing institution from Novartis for advisory board participation and fees paid to employing institution from Proteostasis for advisory board participation, outside the submitted work, and has a patent WO2013061091 pending. Stephen C Hyde also has a patent WO200711062 issued, and a patent WO2013061091 pending. Lee A Davies has a patent US 20140242690 A1 pending, and a patent WO2013061091 pending. Deborah R Gill also has a patent WO200711062 issued, and a patent WO2013061091 pending. James M Wilson is a founder of, holds equity in, an advisor to and grant recipient from, REGENXBIO and Dimension Therapeutics. He is an advisor to Solid Gene Therapy. He is an inventor on patents licensed to various biopharmaceutical companies.

Published July 2016 DOI: 10.3310/eme03050

## **Plain English summary**

Repeated nebulisation of non-viral CFTR gene therapy

Efficacy and Mechanism Evaluation 2016; Vol. 3: No. 5 DOI: 10.3310/eme03050

NIHR Journals Library www.journalslibrary.nihr.ac.uk

# **Plain English summary**

Conly one treatment aimed at the underlying cause is currently available and this is suitable for only the 4% of patients who have specific types of gene mutations; standard care for all other patients targets the symptoms of the disease rather than the cause.

Gene therapy aims to insert a normal copy of the gene back into the lungs' cells and restore function. In the UK CF Gene Therapy Consortium we have (1) identified the best vector with which to gain cell entry; (2) designed a piece of genetic material capable of safely expressing the missing protein for a prolonged duration; (3) found out how we can deliver this to the lungs using a clinical nebuliser; (4) tested safety in two animal models; and (5) confirmed a safe dose in a single-administration trial.

In this trial, CF patients (aged  $\geq$  12 years) randomly received gene therapy or placebo every month for 1 year. The primary outcome was a change in lung health, measured by a standard breathing test [forced expiratory volume in the first second (FEV<sub>1</sub>)]. In addition, we included a number of secondary and safety outcomes.

One hundred and sixteen patients received at least nine doses and were analysed for efficacy. After 12 months, a statistically significant, but modest, difference was observed between the two groups for  $FEV_1$ ; this was supported by trends in other outcomes. Effects were independent of gene mutation, age and sex. There were no significant safety concerns.

We suggest the results reported here provide proof of concept that repeated administration of CF transmembrane conductance regulator (*CFTR*) gene therapy can alter clinically relevant outcomes, providing another step along the path of translational CF gene therapy.

## **Efficacy and Mechanism Evaluation**

ISSN 2050-4365 (Print)

ISSN 2050-4373 (Online)

This journal is a member of and subscribes to the principles of the Committee on Publication Ethics (COPE) (www.publicationethics.org/).

Editorial contact: nihredit@southampton.ac.uk

The full EME archive is freely available to view online at www.journalslibrary.nihr.ac.uk/eme. Print-on-demand copies can be purchased from the report pages of the NIHR Journals Library website: www.journalslibrary.nihr.ac.uk

#### Criteria for inclusion in the Efficacy and Mechanism Evaluation journal

Reports are published in *Efficacy and Mechanism Evaluation* (EME) if (1) they have resulted from work for the EME programme, and (2) they are of a sufficiently high scientific quality as assessed by the reviewers and editors.

#### **EME programme**

The Efficacy and Mechanism Evaluation (EME) programme was set up in 2008 as part of the National Institute for Health Research (NIHR) and the Medical Research Council (MRC) coordinated strategy for clinical trials. The EME programme is broadly aimed at supporting 'science driven' studies with an expectation of substantial health gain and aims to support excellent clinical science with an ultimate view to improving health or patient care.

Its remit includes evaluations of new treatments, including therapeutics (small molecule and biologic), psychological interventions, public health, diagnostics and medical devices. Treatments or interventions intended to prevent disease are also included.

The EME programme supports laboratory based or similar studies that are embedded within the main study if relevant to the remit of the EME programme. Studies that use validated surrogate markers as indicators of health outcome are also considered.

For more information about the EME programme please visit the website: http://www.nets.nihr.ac.uk/programmes/eme

#### This report

The research reported in this issue of the journal was funded by the EME programme as project number 11/14/25. The contractual start date was in March 2012. The final report began editorial review in February 2015 and was accepted for publication in October 2015. The authors have been wholly responsible for all data collection, analysis and interpretation, and for writing up their work. The EME editors and production house have tried to ensure the accuracy of the authors' report and would like to thank the reviewers for their constructive comments on the final report document. However, they do not accept liability for damages or losses arising from material published in this report.

This report presents independent research. The views and opinions expressed by authors in this publication are those of the authors and do not necessarily reflect those of the NHS, the NIHR, the MRC, NETSCC, the EME programme or the Department of Health. If there are verbatim quotations included in this publication the views and opinions expressed by the interviewees are those of the interviewees and do not necessarily reflect those of the authors, those of the NHS, the NIHR, NETSCC, the EME programme or the Department of Health.

© Queen's Printer and Controller of HMSO 2016. This work was produced by Alton *et al.* under the terms of a commissioning contract issued by the Secretary of State for Health. This issue may be freely reproduced for the purposes of private research and study and extracts (or indeed, the full report) may be included in professional journals provided that suitable acknowledgement is made and the reproduction is not associated with any form of advertising. Applications for commercial reproduction should be addressed to: NIHR Journals Library, National Institute for Health Research, Evaluation, Trials and Studies Coordinating Centre, Alpha House, University of Southampton Science Park, Southampton SO16 7NS, UK.

Published by the NIHR Journals Library (www.journalslibrary.nihr.ac.uk), produced by Prepress Projects Ltd, Perth, Scotland (www.prepress-projects.co.uk).

### Efficacy and Mechanism Evaluation Editor-in-Chief

**Professor David Crossman** Bute Professor of Medicine and Dean and Head of Faculty of Medicine, University of St Andrews, and Honorary Consultant Cardiologist, NHS Fife Health Board, UK

### **NIHR Journals Library Editor-in-Chief**

Professor Tom Walley Director, NIHR Evaluation, Trials and Studies and Director of the EME Programme, UK

### **NIHR Journals Library Editors**

**Professor Ken Stein** Chair of HTA Editorial Board and Professor of Public Health, University of Exeter Medical School, UK

Professor Andree Le May Chair of NIHR Journals Library Editorial Group (EME, HS&DR, PGfAR, PHR journals)

Dr Martin Ashton-Key Consultant in Public Health Medicine/Consultant Advisor, NETSCC, UK

**Professor Matthias Beck** Chair in Public Sector Management and Subject Leader (Management Group), Queen's University Management School, Queen's University Belfast, UK

**Professor Aileen Clarke** Professor of Public Health and Health Services Research, Warwick Medical School, University of Warwick, UK

Dr Tessa Crilly Director, Crystal Blue Consulting Ltd, UK

Dr Eugenia Cronin Senior Scientific Advisor, Wessex Institute, UK

Ms Tara Lamont Scientific Advisor, NETSCC, UK

**Professor Elaine McColl** Director, Newcastle Clinical Trials Unit, Institute of Health and Society, Newcastle University, UK

Professor William McGuire Professor of Child Health, Hull York Medical School, University of York, UK

**Professor Geoffrey Meads** Professor of Health Sciences Research, Health and Wellbeing Research and Development Group, University of Winchester, UK

Professor John Norrie Health Services Research Unit, University of Aberdeen, UK

Professor John Powell Consultant Clinical Adviser, National Institute for Health and Care Excellence (NICE), UK

**Professor James Raftery** Professor of Health Technology Assessment, Wessex Institute, Faculty of Medicine, University of Southampton, UK

Dr Rob Riemsma Reviews Manager, Kleijnen Systematic Reviews Ltd, UK

Professor Helen Roberts Professor of Child Health Research, UCL Institute of Child Health, UK

Professor Jonathan Ross Professor of Sexual Health and HIV, University Hospital Birmingham, UK

**Professor Helen Snooks** Professor of Health Services Research, Institute of Life Science, College of Medicine, Swansea University, UK

**Professor Jim Thornton** Professor of Obstetrics and Gynaecology, Faculty of Medicine and Health Sciences, University of Nottingham, UK

**Professor Martin Underwood** Director, Warwick Clinical Trials Unit, Warwick Medical School, University of Warwick, UK

Please visit the website for a list of members of the NIHR Journals Library Board: www.journalslibrary.nihr.ac.uk/about/editors

Editorial contact: nihredit@southampton.ac.uk